Author's response to reviews

Title: Behavioral and psychosocial effects of rapid genetic counseling and testing in newly diagnosed breast cancer patients: Design of a multicenter randomized clinical trial

Authors:

Marijke R. Wevers (m.wevers@nki.nl)
Margreet G.E.M. Ausems (m.g.e.m.ausems@umcutrecht.nl)
Senno Verhoef (s.verhoef@nki.nl)
Eveline M.A. Bleiker (e.bleiker@nki.nl)
Daniela E.E. Hahn (d.hahn@nki.nl)
Frans B.L. Hogervorst (f.hogervorst@nki.nl)
Rob B. van der Luijt (r.b.vanderluijt@umcutrecht.nl)
Heiddis B. Valdimarsdottir (heiddis.valdimarsdottir@mssm.edu)
Richard van Hillegersberg (r.vanhillegersberg@umcutrecht.nl)
Emiel J.Th. Rutgers (e.rutgers@nki.nl)
Neil K. Aaronson (n.aaronson@nki.nl)

Version: 4 Date: 23 December 2010

Author's response to reviews: see over
Dear Editor,

Thank you for reviewing and, in principle, accepting our revised manuscript entitled: "Behavioral and psychosocial effects of rapid genetic counseling and testing in newly diagnosed breast cancer patients: Design of a multicenter clinical trial".

Two last issues were raised during this most recent review by the editor:

1. As requested by the editor, we have added the NutsOhra Foundation as a Competing Interest.
2. As requested, we went through the manuscript formatting checklist and ensured that our manuscript conforms to all of the points. As a result of this, we have added one word in the title of the manuscript and have formatted the reference list, making sure the references contain all named authors.

We hope that improvements are satisfactory and that our manuscript can now be definitely accepted for publication in BMC Cancer.

We look forward to learning of your editorial decision.

Sincerely,

Neil K. Aaronson, Ph.D.
Head, Department of Psychosocial Research
Division of Psychosocial Research & Epidemiology
The Netherlands Cancer Institute

&

Professor, Department of Clinical Psychology
The University of Amsterdam

Plesmanlaan 121
1066 CX Amsterdam
The Netherlands
Tel: +31-20-5122481
Fax: +31-20-5122322
Email: n.aaronson@nki.nl
Website: www.nki.nl